Author's response to reviews

Title: A simple statistical model for prediction of acute coronary syndrome in chest pain patients in the emergency department

Authors:

Jonas Bjork (jonas.bjork@skane.se)
Jakob L Forberg (jakob@hansen.net)
Mattias Ohlsson (mattias@thep.lu.se)
Lars Edenbrandt (lars.edenbrandt@med.lu.se)
Hans Ohlin (hans.ohlin@med.lu.se)
Ulf Ekelund (ulf.ekelund@med.lu.se)

Version: 4 Date: 16 June 2006

Author's response to reviews:

References in text are cited in square brackets and placed before full stop. All authors have been listed in the reference list.

The name of the institutional review board that granted ethical approval for the study has been added to the text.

Figure has been cropped as closely as possible.

Minor editorial changes according to the checklist have been made.